Author’s response to reviews

Title: Extract of Stellera chamaejasme(ESC) inhibits growth and metastasis of human hepatocellular carcinoma via regulating microRNA expression

Authors:

Xiaoni Liu (lxnlxm@126.com)
Shuang Wang (wsh1029@126.com)
Jianji Xu (15088462963@163.com)
Buxin Kou (koubuxin@sina.com)
Dexi Chen (dexichen@ccmu.edu.cn)
Yajie Wang (wyj056@126.com)
Xiaoxin Zhu (zhuxx59@163.com)

Version: 3 Date: 30 Jan 2018

Author’s response to reviews:

Dear editor:

Thank you for your kindly information!

I have corrected the manuscript according to your requirments.

The responses point to point are as follows:

1. Please detail in your Methods section the euthanization method used in your study.

The euthanization method has been added.

2. Please provide all authors in the submission system, where only author X. Liu is indicated.

I have provided all authors in the submission system.
3. Please amend the headings of the abstract (Background; Methods; Results; Conclusions) and/or of the manuscript (Background; Methods; Results; Discussion; Conclusions) in accordance to BMC CAM formatting.

I have amended the headings of the abstract and/or of the manuscript.

4. Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

A statement “all authors have read and approved the manuscript” has been included in the Authors' contributions section.

5. A deposition number of the plant should be included, if available.

A deposition number has been added.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

I have uploaded a clean manuscript in the system.

Thanks again!

Best regards!

Liu xiaoni